## University of Minnesota

Twin Cities Campus

Sarah M Greising, PhD

Assistant Professor McKnight Land-Grant Professor

School of Kinesiology

College of Education and Human Development

220A Cooke Hall 1900 University Ave SE Minneapolis, MN 55455

Phone: 612-626-7890 Fax: 612-626-7700 Email: gre0064@umn.edu

16. June. 2021 Dear Dr. Saha,

We would like to thank you for the editorial comments for our manuscript (JoVE62905) entitled "In Vivo Measurement of Hindlimb Dorsiflexor Isometric Torque from Pig".

The specific responses to all comments are noted in the tracked version of the manuscript. All comments were noted as "completed". On behalf of my co-authors, we look forward to the publication of our manuscript. Please do not hesitate to contact me if there are any issues related to this revision that you would like to discuss.

Sincerely, Sarah Greising, PhD